Author’s response to reviews

Title: The Invested in Diabetes Study Protocol: A Cluster Randomized Pragmatic Trial Comparing Standardized and Patient-Driven Diabetes Shared Medical Appointments

Authors:

Bethany Kwan (Bethany.kwan@cuanschutz.edu)
L. Dickinson (miriam.dickinson@cuanschutz.edu)
Russell Glasgow (russell.glasgow@cuanschutz.edu)
Martha Sajatovic (martha.sajatovic@uhhospitals.org)
Mark Gritz (mark.gritz@ucdenver.edu)
Jodi Holtrop (jodi.holtrop@cuanschutz.edu)
Donald Nease, Jr. (donald.nease@cuanschutz.edu)
Natalie Ritchie (natalie.ritchie@dhha.org)
Andrea Nederveld (andrea.nederveld@cuanschutz.edu)
Dennis Gurfinkel (dennis.gurfinkel@cuanschutz.edu)
Jeanette Waxmonsky (jeanette.waxmonsky@cuanschutz.edu)

Version: 2 Date: 19 Nov 2019

Author’s response to reviews:

Along with changes made in the last revision, we have changed ICC from percents (3% and 5%) to decimals (0.03 and 0.05)